This is a point by point reply to reviewer’s comment

Specific Comments to Authors: Dear Sir, I like your editorial, however, it seems to be confusing that you should modified the title as the objective is to evaluate the use of an oral solution for IBS-diarrhea patients, even thought you also mention its probable use in other organic causes of chronic diarrhea. As well, it could be useful if you could give some information, if it exist to functional chronic diarrhea, for example and how it could be of beneficial in each case, or not. Hope you could help me out with these ideas.
Sincerely

Thanks for the reviewer's valuable suggestions. As suggested we have changed the title so as not to create confusion